Author's response to reviews

Title: What happens to ART-eligible patients who do not start ART? Dropout between screening and ART initiation: a cohort study in Karonga, Malawi

Authors:

Nuala M McGrath (nuala.mcgrath@lshtm.ac.uk)
Judith R Glynn (judith.glynn@lshtm.ac.uk)
Jacqueline Saul (jacky.saul@lshtm.ac.uk)
Katharina Kranzer (katharina.kranzer@lshtm.ac.uk)
Andreas Jahn (drandreasjahn@gmail.com)
Frank Mwaungulu (mia.crampin@lshtm.ac.uk)
Msenga HC Ngwira (mia.crampin@lshtm.ac.uk)
Hazzie Mvula (mia.crampin@lshtm.ac.uk)
Fipson Munthali (mia.crampin@lshtm.ac.uk)
Venance Mwinuka (mia.crampin@lshtm.ac.uk)
Lorren Mwaungulu (mia.crampin@lshtm.ac.uk)
Paul EM Fine (paul.fine@lshtm.ac.uk)
Amelia C Crampin (mia.crampin@lshtm.ac.uk)

Version: 5 Date: 12 October 2010

Author's response to reviews:

On page 14, I changed 'the proportion of individuals in WHO stage 4 at screening (64%)' to 'the proportion of individuals in WHO stage 4 at screening (55%)', and I added one reference to the bibliography (#21) which resulted in the remaining references being renumbered.